\documentclass{article}
\bibliographystyle{plain}
\begin{document}
\title{Typos in Chapter 17}
\date{ (Revised \today)}

\maketitle

\begin{enumerate}
\item
On page 285, the displayed equations between (10.4.28) and (10.4.30) should be
labelled (10.4.29). 
\item
         On page 287, in the description of the Predictor-corrector method, after
"... $\tau \in (0,1/15)$ be given.", insert: Assume that
         $(X_0,y_0,S_0) \in {\cal N}_F(\tau)$. 
\item
         On page 290, line 14 should start: Using the first two equations of
(10.4.29) and the fact ... 
\item
         On page 297, Theorem 10.4.6 should be attributed to Monteiro and Zhang
[547]. 
\end{enumerate}

\bibliography{.master,.psd,.publs}
\end{document}

